Author’s response to reviews

Title: A case of posterior reversible encephalopathy syndrome secondary to asymptomatic post streptococcal glomerulonephritis in a child with sickle cell anemia: a case report.

Authors:

ehab Hanafy (ehab.hmahmoud@gmail.com)
Duaa Alshareef (wrdha2012@hotmail.com)
Suhaila Osman (suhailasalah@hotmail.com)
Abdullah Al Jabri (Dr.aaaljabri@gmail.com)
Faisal Nazim (nazim_prof@yahoo.com)
Gihan Mahmoud (drgihanmahmoud@gmail.com)

Version: 2 Date: 15 Dec 2017

Author’s response to reviews:

Thank you for your valuable comments an sorry for being late.

- I have added a paragraph at the end of introduction in the abstract area mentioning the rarity of this case association in children specifically and why this condition should be interpreted carefully in SCA population.

- Lab works done which showed high anti streptolysin O titer and significantly low complement levels with RBCs in urine plus picture of hypertension and these are all mentioned within the text and I think they all point to post streptococcal glomerulonephritis.

- A full paragraph including medical history, family history and history regarding medications, allergies and surgeries was added at the beginning of the case presentation section

- Physical and neurologic examination together with vital signs at presentation were added in the second paragraph in the case presentation section

- Acute chest syndrome and vaso-occlusive crisis were discussed with more details and justified in the second paragraph in the case presentation section
• All the relevant lab works done for the patient is in the table (table 1) and other lab works i.e. ASO titer, C3, urine analysis, renin and aldosterone are within the text – paragraph 7 in case presentation section

• The case was summarized at the beginning of the discussion and besides the rarity of the condition, the message was mentioned in the discussion and also in the conclusion section.

I would like to thank you again for your comments hoping that this review will meet your expectations.